Author's response to reviews

Title: BMI and Mental Ill-Health in 68,000 Swedish Adults. Controlling for an age-gender combination.

Authors:

Susanne Brandheim (susanne.brandheim@kau.se)
Ulla Rantakeisu (ulla.rantakeisu@kau.se)
Bengt Starrin (bengt.starrin@kau.se)

Version: 2 Date: 20 June 2012

Author's response to reviews:

Changes:
Added details on ethical approval under Method
Added a section on competing interests
Added a section on Authors’ contributions